Author’s response to reviews

Title: Use of a trigger tool to detect adverse drug reactions in an emergency department

Authors:

Silvana Almeida (silvana.mariadealmeida@gmail.com)
Aruana Romualdo (aruana.romualdo@einstein.br)
Andressa Ferraresi (andressa.ferraresi@einstein.br)
Giovana Zelezoglo (giovana.zelezoglo@einstein.br)
Alexandre Marra (alexandre-rodriguesmarra@uiowa.edu)
Michael Edmond (michael-edmond@uiowa.edu)

Version: 3 Date: 18 Oct 2017

Author’s response to reviews:

Answers to the Editor

It has been assessed by our reviewers and based on these reports, I am pleased to inform you that it is potentially acceptable for publication in BMC Pharmacology and Toxicology, once you have carried out some minor editorial revisions:

Authors: We really appreciate your comments.

1. Please include a title page to your manuscript that is properly formatted according to our submission guidelines (https://bmcpharmacoltoxicol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article). The title page should:

   - present the title that includes, if appropriate, the study design
   - list the full names, institutional addresses and email addresses for ALL authors
   - indicate the corresponding author

Authors: We have included it.
2. Please format your abstract to include only the following subheadings: Background, Methods, Results, Conclusions.

The current 'Objectives' section can be incorporated into the Background.

Authors: We have changed it.

3. Please provide more justification for the contributions of authors GRZ, ME, and ARM, as currently their contributions do not automatically qualify for authorship. Please amend your manuscript accordingly. According to the ICMJE guidelines, to qualify as an author one should have:

- made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data;

- been involved in drafting the manuscript or revising it critically for important intellectual content;

- given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; and

- agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Acquisition of funding, collection of data, or general supervision of the research group, alone, does not usually justify authorship.

Authors: We have clarified it.
4. Please note that the Availability of data statement should detail the location of the raw data generated or analysed during this study. We note that only summarized data is presented in this manuscript, so we would like to ask that you amend this statement. We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository. If there are any ethics restrictions preventing the sharing of the raw data, please state this. If you do not wish to share the data, please include a reason why it will not be shared in this section.

For more information and a list of suitable availability statements, please see our Submission guidelines: https://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials

Authors: Confidentiality agreements prevent us from sharing the raw data generated during this study.

5. Please be sure to cite 'Figure 1' in the text of your manuscript to avoid removal by our production department.

Authors: We have cited it.

6. Please cite and indicate where the tables should appear at the relevant locations in the text file so that the tables can be added in the correct place during production.

Authors: We have cited it.

7. Upon uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

Authors: We have removed any tracked changes.